


e SSand AlS Shaft/Rope Inspections: The weekly MSHA compliance inspection for the AIS is planned for
today. The inspection of the SS was completed on 4/3.

e Contamination survey to the waste face (Phase 3): This is being broken into multiple activities. The first activity
will be surveying down to $2150 at E140. The safety documents have been submitted to DOE and approval to
complete the first activity may occur late this week.

e Replacement HEPA filters are due to arrive starting this week. The filter change mock-up is planned for the week
of April 14th. NWP has identified that they will be task ready to change out the filters by 5/22.

e Shipments of TRU waste from LANL to WCS are continuing. EPA is visiting WCS tomorrow.

e NWP has provided a recommendation to discontinue future salt sales.

e Pictures from the entry last week have been posted on the Portal

e Shaft Inspections tomorrow

Attached are two charts showing HEPA flow rates and Waste Hoist Tower DPs from 2/3-2/15. The color code is as
follows:
e Cyan =1 Pre-Filter MOD
Green = 2 Pre-Filter HIGH
Yellow = 1* HEPA Bank
Red = 2"* HEPA Bank

NMED asked about filter change length? WIPP built mock up and are doing walk trough’s — will probably revise
estimate. SRS experts will be working on this and will have a better estimate of time.

EPA Nick Stone: Trigger is 2.5 and is at 2.0 — Are they confident that will function during re-entry? WIPP: Will change
mod, high and HEPA’s on one unit and only mod and high on other. Also looking at adding another HEPA bank. If mod
filters reached alarm at 2.5, they might have to cancel UG entry. If increase continues at current rate, they have about 3
weeks. People in UG will have PPE (personal protective equipment) so the danger is not immediate to them. Looking at
adjusting flows so that burden on clogging filter is lessened. This may extend the three week estimate.

New action item: Will provide waste disposition map and list of waste in Panel 7 Room 7.

NMED: Still curious about set point for CAMs. Why isn’t 30 the first choice for Station B? Why 40 DAC hour? What are
set points for CAMS in UG? High is 30, High High is 50. Not connected to CMR (local alarms).

EPA (Kathy Economy): Looking at active disposal room flow rate normal is 50,000 cfm. What will it be in filtration
mode? NWP: Question is about active room. These rooms are not active right now. Will stay in filtration mode for
some time. lJill Farnsworth will say more on tomorrow’s call.

EPA (Tom Peake): Joe Franco said there will be Rad incident Report out? Oba: Report is almost ready, thinking it will
come out next week. Will be about emergency response, ventilation adequacy ~ not identified as critical nuclear safety,
things from fire report, about 160 pages....

EPA{ n Walsh): Have seen or will shortly see filter changes and equipment at most sampler sites.

Oba: Checking in that frequency is still OK. Proposes that once they see the waste face, they would like to reduce the
frequency.

As a reminder, the call-in number is (866) 723-6758, code 8040901#

Thanks



Oba























































































































